Author’s response to reviews

Title: Assessment of patient safety culture in primary health care in Muscat, Oman: A questionnaire-based survey

Authors:
Muna Habib AL Lawati (drmunali@gmail.com)
Stephanie Short (stephanie.short@sydney.edu.au)
Nadia Noor Abdulhadi (nadianoorah@yahoo.com)
Sathiya Murthi Panchatcharam (sathiya.m@omsb.org)
Sarah Dennis (sarah.dennis@sydney.edu.au)

Version: 1 Date: 02 Feb 2019

Author’s response to reviews:

Dear Editor BMC Family Practice,


Thank you for sending this manuscript for review. The editor's comments have been very useful and we have revised the manuscript accordingly.

Thanking you,

Muna Habib